Author's response to reviews

Title: Is the arch form influenced by sagittal molar relationship or Bolton tooth-size discrepancy?

Authors:

Abdullah M Aldrees (abdullah@post.harvard.edu)
Abdulmajeed M Al-Shujaa (o_ams_o@hotmail.com)
Mohammad A Alqahtani (malqahta@buffalo.edu)
Ali S Aljhani (aljhani.a@gmail.com)

Version: 5 Date: 30 March 2015

Author's response to reviews: see over
Dear Editor-in-Chief,

We would like to thank you for the meticulous evaluation of the submitted manuscript and the valuable comments.

- **“please place reference # after each author”**
  The recommended correction has been made, and the following reference numbers appeared after each author:
  
  Whereas Sameshima [10], Araujo and Souki [4], Akyalcin et al. [11], Basaran et al. [12], Nie and Lin [3], Al-Tamimi and Hashim [13], and Endo et al. [14] all found no significance when relating sex to tooth-size ratios.

- **“please consider changing 'patients' into 'ethnicity'”**
  The recommended correction has been made, and the word “ethnicity” was used.

- **“please add per cent values in table 3”**
  The recommended correction has been made, and the percentages were added to Table 3.

- **“please add per cent values above each column in the graph (figure 1)”**
  The recommended correction has been made, and the percent values were added above each column in Figure 1.